Author's response to reviews

Title: Successful treatment of a schizophrenic patient with phenotypic evidence for CYP2D6 ultrarapid metaboliser status and non-response to amisulpride with melperone-augmented haloperidole: a case report

Authors:

Maximilian Gahr (maximilian.gahr@uni-ulm.de)
Regina Gastl (regina.gastl@uni-ulm.de)
Markus M Kölle (markus.koelle@uni-ulm.de)
Carlos Schönfeldt-Lecuona (carlos.schoenfeldt@uni-ulm.de)
Roland W Freudenmann (roland.freudenmann@uni-ulm.de)

Version: 2 Date: 18 August 2011

Author's response to reviews:

Dear ladies and gentlemen,

formal aspects of the manuscript were changed accordingly.

Sincerely,

Maximilian Gahr